ORIGINAL ARTICLE

Revised: 18 August 2022

WILEY

Anti-osteoporosis medication dispensing by clinical commissioning groups in England – an ecological study of variability in practice and of the effect of the Covid-19 pandemic

Sobia S. Janjua¹ | Helen F. Boardman¹ | Arvind Sami² | Antony Johansen^{3,4} | Li Shean Toh¹ | Kassim M. Javaid^{2,4}

¹Division of Pharmacy Practice and Policy, School of Pharmacy, University of Nottingham, Nottingham, UK

²Nuffield Department of Orthopaedics, Rheumatology, and Musculoskeletal Sciences, Botnar Research Centre, University of Oxford, Oxford, UK

³School of Medicine, University Hospital of Wales, Cardiff University, Wales, UK

⁴Falls and Fragility Audit Programme, Royal College of Physicians, London, UK

Correspondence

Li Shean Toh, Division of Pharmacy Practice and Policy, School of Pharmacy, University of Nottingham, University Park, Nottingham NG7 2RD, UK.

Email: lishean.toh@nottingham.ac.uk

Abstract

Purpose: To investigate whether the rate of Anti-Osteoporosis Medication (AOM) dispensing was related to prevalence of risk factors and hip fracture incidence in the local population.

Methods: The Open Prescribing database was used to analyse dispensed AOM at the level of Clinical Commissioning Groups (CCGs) in England. Male Healthy Life Expectancy (MHLE), Female Healthy Life Expectancy (FHLE), the prevalence of smoking and active adults, the incidence of hip fracture and of alcohol related hospital admissions, and local dispensing of a comparator drug (atorvastatin) were considered as predictor variables. Linear and multilinear regression were performed. Using atorvastatin as a comparator, AOM dispensing was compared after the start of the Covid-19 pandemic with the same quarter the previous year.

Results: Rates of AOM per 1000 people aged over 65 years in a CCG area varied between 379.2 and 1129.1, with a mean of 670.3. Population risk factors were individually related to the amount of AOM dispensed in an area. Collectively, local activity levels in adults (p = 0.042) and local hip fracture incidence (p = 0.003) were significantly negatively correlated with rates of AOM dispensed. Rates of alendronate dispensing fell significantly at the start of the Covid-19 pandemic (p < 0.001), whilst atorvastatin dispensing rates significantly increased (p < 0.001).

Conclusion: Lower rates of AOM dispensing were seen in areas with a higher proportion of active adults and higher incidence of hip fracture. Multidisciplinary services should be developed to address this care gap with consideration given to local population risk factors. Community pharmacists are ideally placed to play a vital role in osteoporosis management.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made. © 2022 The Authors. *Pharmacoepidemiology and Drug Safety* published by John Wiley & Sons Ltd.

1

KEYWORDS

bisphosphonates, community pharmacy, COVID-19, denosumab, osteoporosis medication, real-world data

Plain language summary

Osteoporosis leads to a "thinning" of the bones which makes them brittle and more likely to fracture. However, effective medications are available which prevent and treat the condition. Dispensing of medication for osteoporosis varies throughout England and fell during the first 3 months of the Covid-19 pandemic. Areas with higher hip fracture cases had lower rates of osteoporosis medication dispensed, indicating more people at risk need to start medication to prevent hip fractures occurring. Furthermore, areas where there were higher proportions of active adults had lower rates of osteoporosis medication dispensed, indicating a reduced need and highlighting the potential an active lifestyle could play in preventing osteoporosis.

Key Points

- Osteoporosis results in weaker bones, which increases risk of fracture, of which hip fractures are the most serious.
- The amount of anti-osteoporosis medication dispensed in England in 2019 was analysed against local population risk factors for the condition.
- The Covid-19 pandemic saw a significant decrease in the amount of alendronic acid dispensed during the first quarter after the national lockdown in England.
- Hip fracture incidence and adult activity levels were significantly negatively related to rates
 of anti-osteoporosis medication dispensed.
- Initiation of treatments to prevent hip fractures and promotion of active lifestyles may improve osteoporosis management and prevention in England.

1 | INTRODUCTION

Osteoporosis is a long-term condition characterised by weakened microarchitecture and strength of bones.¹ Its asymptomatic nature² means that it is often only diagnosed after a fragility fracture,³ which follows low level trauma such as a fall from a standing height. Hip fractures are one of the most severe outcomes of osteoporosis. They reduce mobility, quality of life⁴ and there is a three-fold increase in risk of death after a hip fracture.⁵ However, only one fifth of the deaths after a hip fracture can be explained by the fracture, with comorbidities accounting for the rest.⁶

Costs related to hip fracture have been reported as over £16 000 per patient, including hospital stay and two-year post fracture care.⁷ Risk factors for osteoporosis include history of fragility fracture, untreated premature menopause, low BMI (<18.5 kg/m²), smoking, alcohol intake (>14 units/week), family history of hip fracture and frequent use of glucocorticoids.⁸

Annual incidence for osteoporotic hip fractures has been predicted to affect 6.26 million people worldwide by 2050.⁹ Gender influences fracture rates and it has been estimated that 1 in 3 women¹⁰ and 1 in 5 men¹¹ will suffer a fragility fracture after the age of 50 years.¹² In the United Kingdom, osteoporosis effects over 3 million people. Annually, this results in more than half a million people receiving treatment in hospital for fragility fractures.¹² These figures highlight osteoporosis as a significant health problem which needs to be addressed.

Fortunately, effective treatments are available which significantly reduce fracture risk evidenced by both randomised clinical trials¹³ and real-world evidence.¹⁴ In the United Kingdom, treatments for osteoporosis are listed in The British National Formulary (BNF)¹⁵ and recommended in National Institute of Clinical Excellence (NICE) guidance. These include bisphosphonates,¹⁶ denosumab,¹⁷ raloxifene,^{18,19} and parathyroid hormone analogues.¹⁹

Populations with higher prevalence of risk factors have more people appropriate for Anti-Osteoporosis Medication (AOM), and in a needs-based healthcare system capacity and capability should reflect this.²⁰ In England, social deprivation has been associated with higher hip fracture incidence.²¹ However, there is a lack of literature examining whether risk factors associated with osteoporosis and social deprivation, for example, healthy life expectancy, smoking, activity levels and excessive alcohol intake may be related to rates of AOM use. Furthermore, whether consequences such as hip fractures are related to AOM use.

We investigated the relationship between osteoporosis risk factors and incidence of hip fracture with the rate of AOM dispensing at the regional level in England.

The Covid-19 pandemic has affected fracture rates,²² and the management of osteoporosis, including bone health assessments²³ and the delivery of bone density scans,²⁴ so our study also considered

variation in AOM dispensing in England, both before and after the start of the pandemic.

2 | METHODS

2.1 | Study design and setting

This ecological study used the Open Prescribing database to analyse the number of products of AOM dispensed via community pharmacies in England in 2019.²⁵ The effect of the Covid-19 pandemic was investigated by comparing the first quarter after the national lockdown in England (April till June 2020) with the same quarter the previous year (April till June 2019).

2.2 | Participants and study size

Data were extracted for Clinical Commissioning Group (CCG) areas in England. A CCG covers a geographical area, where GPs and nurses have more input into services and allocation of costs.²⁶ Over time CCG areas have merged to provide improved healthcare to local populations. Dispensed AOM, risk factors and outcomes were obtained for 135 CCG areas in 2019.

2.3 | Data sources

2.3.1 | The open prescribing database

The Open Prescribing database²⁵ contains information from the NHS Business Service Authority's monthly published files. It lists the month, number of products, quantity of units and actual cost for individual medications dispensed via community pharmacies or personally administered by doctors in England from July 2015 until present. The database allows for analysis at the CCG level and allocates codes for individual medication products as listed in the BNF.¹⁵ We included medications listed in NICE guidance¹⁶⁻¹⁹ (see Supplementary Table 1). Monthly AOM items dispensed at the CCG level were added to obtain the annual items for each CCG in 2019.

2.3.2 | The Office for National Statistics

The Office for National Statistics (ONS) lists the number of people of every age at mid-year 2019 in a CCG area,²⁷ enabling calculation of the total number of people aged over 65 years. Product quantities and population numbers were used to calculate the outcome variable of AOM dispensed per 1000 people aged over 65 years.

3

England is also divided into Health Authority (HA) areas which administer the National Health Service. The ONS holds local population risk factor data for HA areas. Risk factors associated with osteoporosis were extracted and prevalence compared to dispensed AOM. Clinical Commissioning Group areas were mapped to Health Authority (HA) areas by population numbers from the ONS.²⁸

2.3.3 | The National Hip Fracture database

The number of people in a CCG area who sustained a hip fracture were extracted from the National Hip Fracture Database for 2019.²⁹

2.4 | Inclusion criteria

Osteoporosis medication items dispensed via community pharmacies in England and recommended in NICE guidance (bisphosphonates¹⁶: alendronate, risedronate, ibandronate, and non-bisphosphonates¹⁷⁻¹⁹: denosumab, raloxifene, teriparatide and strontium ranelate) were analysed. Where more than one indication existed for a drug, the dose specific preparation for osteoporosis was used. Raloxifene was considered an AOM since its other indication (breast cancer prevention) is not licensed in the United Kingdom.

2.5 | Exclusion criteria

Bone medications listed in the BNF section 6.6 which are not indicated for osteoporosis and not included in NICE guidance for osteoporosis were excluded, in addition to formulations / doses not used solely for osteoporosis. Zoledronic acid was excluded due to negligible dispensing via community pharmacy. Calcium and/or vitamin D were not included, since their effectiveness in this context is primarily as a co-prescription with the AOM listed above. Clinical Commissioning Group areas were excluded where there was a greater than 5% population difference with the corresponding HA. Also, if there was more than one CCG in an HA area and if population risk factors were not available at the CCG level.

2.6 | Variables

We examined the following ecological level osteoporosis risk factors from the ONS: Male Healthy Life Expectancy (MHLE), Female Healthy Life Expectancy (FHLE), percentage of people who smoke, percentage of active adults (at least 150 minutes of moderate intensity physical activity per week) and rate of alcohol related hospital admissions per 100 000 population.³⁰ Weighted averages were taken for HA areas which had merged, with less than 5% population discrepancy with corresponding CCG area.

2.7 | Covid-19 pandemic

The most commonly dispensed oral and injectable AOMs were identified and compared to atorvastatin, the most commonly dispensed medication in England for hyperlipidaemia; another asymptomatic long-term condition whose prevalence increases with age (see Supplementary Table 2, for codes).³¹

Atorvastatin was therefore used as a comparator drug.

2.8 | Statistical methods

Data was analysed using IBM SPSS statics V28.³² Outcomes and predictors were described using descriptive parametric and nonparametric statistics. Linear and multilinear regression methods were used to analyse population level osteoporosis risk factors on AOM dispensing rates per 1000 adults aged over 65 years in each region. Assumptions of a regression analysis model were checked by Cook's distance to confirm extreme outliers.

To investigate the early effect of the Covid-19 pandemic, the rates dispensed were compared for the first quarter (April – June 2020) after the national lockdown in England (23rd March 2020) to the same quarter the previous year (April – June 2019). Due to CCG mergers, mid-year population data for 2020 was presented for 106 CCGs,³³ therefore, to enable comparison, population data from 2019²⁸ was added for mergers.³⁴ Two-tailed p values <0.05 were considered significant.

3 | RESULTS

Some CCG areas had incomplete data (see Supplementary table 3 for exclusions and missing data). The characteristics for all CCGs and those with complete predictor data are compared in Table 1. The mean CCG population size was 416 940 and the mean proportion aged 65 years and over in a CCG area was 18.5% (range from 6.4% to 28.3%). The total AOM dispensed via community pharmacies in England in 2019 ranged from 10 257 to 221 398 products for individual CCG areas. For every 1000 people aged over 65 years, there were between 379 and 1129 AOM products dispensed at the CCG level.

3.1 | Linear regression

For CCGs with complete data (n = 94), simple linear regression showed there were significant relationships (p = <0.05) between MHLE, FHLE, percent smoked, percent active adults and alcohol related hospital admissions per 100 000 with the rate of AOM dispensed per 1000 people aged over 65 years.

TABLE 1 Characteristics of Clinical Commissioning Groups (CCGs) in 2019 Commissioning Groups

Characteristic	All CCGS (n = 135)	CCGs with complete data (n = 94)	
Total population ^a in England	56 286 961	42 354 674	
Mean population ^a in CCG areas	416 940	450 582	
Total population aged 65 years and over ^a in England	10 353 716	7 595 908	
Mean population aged 65 years and over ^a in CCG areas	76 694	80 808	
Mean proportion aged 65 years and over in CCG areas (%) ^a	18.5	17.7	
Mean proportion female aged 65 years and over in CCG areas (%) ^a	10.1	9.6	
Mean male healthy life expectancy ^a in CCG areas (years)	X	62.0	
Mean female healthy life expectancy ^a in CCG areas (years)	X	62.5	
Mean Smoked ^a in CCG areas (%)	Х	18.0	
Mean Active adults ^a in CCG areas (%)	х	56.0	
Mean Alcohol-related hospitalisation admissions per 100000ª in CCG areas	х	672	
Total AOM dispensed ^c in England	6 777 429	54 628	
Mean AOM per 1000 aged over 65 years ^{a,c} in CCG areas	670	701	
Hip fracture cases ^b in England	54 343	39467 ^d	
Mean atorvastatin per 1000 ^c aged over 65 years ^{a,c} in CCG areas	4850.5	5246	

^aOffice for National Statistics.

^bHip Fracture Database.

^cOpen Prescribing database.

 $^{\rm d}n=93$ (missing data), X = excluded (See Supplementary Table 3 for exclusion details).

3.2 | Multilinear regression model

Due to exclusions (see Supplementary Table 3), 93 CCGs were included in the multilinear regression model (see Figure 1). The independent predictors of AOM rates per 1000 adults aged 65 years and over are shown in Table 2. Thirty one percent of the variation in AOM rate per 1000 aged over 65 years was explained by the population characteristics, predictors in the model ($p \le 0.001$, $R^2 = 0.31$). An increase of 100 hip fracture cases in a CCG area was significantly associated with a decrease of 22 items in the AOM dispensing rate. As the average population aged over 65 years in a CCG was 766,94, an increase in active adults by 1% significantly reduced the dispensing rate of AOM per 1000 aged over 65 years by an average of 77 per CCG area.

3.3 | Effect of Covid-19 pandemic on individual types of AOM dispensed

Pre-pandemic (2019), the most commonly dispensed oral AOM was Alendronate (5 912 234 products) and the most commonly dispensed injectable AOM was Denosumab (61 010 products). Excluding 3 CCGs with no community dispensing of Denosumab, the ratio of Denosumab versus Alendronate dispensed varied from 0.01% to 9.6% (median 0.5%). During the first quarter after the start of the pandemic (April to June 2020), rates dispensed of Alendronate per 1000 people aged over 65 years decreased across 89.6% (n = 95) of CCGs, when compared to the same quarter in 2019 ($p \le 0.001$). Rates of Denosumab also decreased across



FIGURE 1 Selection of CCGs for multilinear regression. A = greater than 5% population difference in Clinical Commissioning Group (CCG) and corresponding Health Authority (HA) area. This includes single HA to a CCG area and where more than one HA covers a CCG area. B = More than one CCG in HA area. C = HA risk factor data exists for previous boundaries. D = Missing data. 39.6% (n = 42) of CCGs but were the same for 9.4% (n = 10) of CCGs with no overall significant change (p = 0.12). In contrast, all CCG's in England had an increase in the dispensed rates of atorvastatin in the quarter April to June 2020 (n = 106), compared to the same quarter in 2019 ($p \le 0.001$).

4 | DISCUSSION

Dispensed AOM rates varied across CCGs, indicating disparities in prescribing and subsequent dispensing behaviour. Much of this variation was not explained by ecological level predictors, and indeed ran counter to what these might suggest to be appropriate.

Improving effective osteoporosis management is a priority for the NHS, in both primary and secondary prevention settings.³⁵ Identification of populations with highest prevalence of risk factors for osteoporosis treatment would help to tailor interventions to local needs.

However, we have demonstrated counter-intuitive associations between rates of AOM amongst people aged over 65 years and both the local incidence of hip fracture and the local prevalence of adult activity. Accepting the limitation of ecological analyses this suggests a mismatch between local health-care need and delivery for osteoporosis.

4.1 | Local Hip fracture incidence

The introduction in 2005 of NICE guidelines on osteoporosis treatments to prevent fractures and the availability of generic alendronic acid has been associated with a significant increase in prescribing of osteoporosis medication and a reduction in hip fracture incidence.³⁶ Our study found that in 2019, hip fracture incidence was higher in areas with lower rates of AOM dispensed. This highlights the potential impact greater availability of osteoporosis treatments could have on prevention of hip fractures.

TABLE 2Univariate and multivariate regression showing the effect of a one unit increase in predictor (Unstandardised B) on the rate of Anti-
Osteoporosis Medication (AOM) dispensed per 1000 people aged over 65 years

		Univariate	Univariate		Multivariate	
Predictor	n	В	95% CI for B	n	В	95% CI for B
Male healthy life expectancy ^a (years)	94	-16.4	-25.0 to -7.9	93	7.5	-13.5 to 28.4
Female healthy life expectancy ^a (years)	94	-17.8	-25.8 to -9.8	93	-14.9	-34.1 to 4.2
Smoked ^a (%)	94	15.6	6.0 to 25.2	93	0.5	-13.7 to 14.7
Active adults ^a (%)	94	-11.4	-17.3 to -5.5	93	-7.2	-14.2 to -0.3
Alcohol-related hospital admissions per 100,000 ^a	94	0.3	0.1 to 0.5	93	0.1	-0.2 to 0.4
Hip fracture cases / 100 ^b	134	-4.1	-11.3 to 3.2	93	-21.6	-35.9 to -7.4
Total atorvastatin/1000 ^c	135	0.1	-0.3 to 0.2	93	0.4	0.2 to 0.6

^aOffice for National Statistics.

^bHip Fracture Database.

^cOpen Prescribing database (See Supplementary Table 3 for exclusion details), n = number of CCG areas.

4.2 | Local activity levels

Results of a systematic review found physical activity probably prevents osteoporosis in people aged 65 years and over.³⁷ Our study strengthens this finding as the percent of active adults in areas of England is both individually and collectively significantly related to lower rates of AOM dispensed, and therefore indicative of prevalence of osteoporosis in the local population. A 1% increase in active adults was associated with a reduction in approximately seven AOM items per 1000 people aged over 65 years. This was significant (p = 0.042) and highlights the importance of keeping active to prevent osteoporosis. However, other factors such as socioeconomic deprivation add to the complexity of the situation.

4.3 | Local socio-economic deprivation

The Index of deprivation for an area considers aspects such as employment, education levels, health and living conditions of the population.³⁸ Although our study did not specifically look at deprivation, our predictor variables have been linked to deprivation. Lifestyle risk factors such as smoking.³⁹ and alcohol consumption, in particular binge drinking⁴⁰ are more prevalent in deprived areas. Although, women are more likely to suffer from osteoporosis.¹⁰ those in the least deprived quartile of the population had significantly higher heel Bone Mineral Density (BMD) compared with the rest of the population.⁴¹ Women in this quartile were also significantly less likely to smoke, indicating a relationship between gender, smoking, deprivation and osteoporosis. Socioeconomic deprivation was also found to be a risk factor for osteoporosis in older adults.⁴² Our study did not include screening BMD and participants, however, the amount of medication dispensed in a CCG area was used as an indication of scale of occurrence of osteoporosis. Therefore, establishing the amount of medication people take in an area could be used to develop potential interventions to reduce disparities and improve lifestyles.

4.4 | Community pharmacy and the Covid-19 pandemic

Whilst rates of atorvastatin dispensing increased during the Covid-19 pandemic, there was a significant reduction in dispensing of alendronate and no change in that of denosumab.

The data presented in the Open Prescribing database contains the number of items of medication dispensed via community pharmacies in England. Analysis of the database highlights the variation and scope of interaction that people with osteoporosis have with the pharmacists and the potential for improve pharmacy-based osteoporosis services. This could include identification of people at risk for osteoporosis. Furthermore, the interaction at the pharmacist in a unique position to deliver an intervention that may improve osteoporosis management. A continuous and tailored patient-centred education on importance of keeping active or identification of any medication related concerns which prevent adherence is recommended. The impact of this interaction was highlighted during the Covid-19 pandemic when rates of the most commonly dispensed oral AOM, alendronate, was reduced. The picture was slightly different for denosumab with supplies not as affected. Denosumab is administered six monthly and NICE recommended treatment should not be postponed due to the Covid-19 pandemic.⁴³ Continued supply via community pharmacy enabled NICE guidelines to be met. This in part is due to their accessibility, as in England, 89% of the population live within a 20-minute walk of a community pharmacy, rising to 99% in deprived areas.⁴⁴

4.5 | Strengths and limitations

These analyses used real world data from across England to describe the pattern of AOM use at the regional level. However, the data sources; Open Prescribing database, National Hip Fracture Database and the Office for National Statistics are unlinked. Although being able to link the data would be helpful, this is not possible, therefore considered when making associations. The major limitation of this methodology is the lack of patient level information which precludes assumptions of causality.⁴⁵

The AOM data analysed was for 2019, but data sources of population risk factors were not all from the same calendar year. However, they are based on the latest Public Health England and the National Population Survey data available at the time of analysis for Health Authority areas in 2017.³⁰

Population discrepancies of up to 5% between CCG and HA were included, however, these could equate to large numbers. Dispensed item numbers do better reflect patient use of AOM than prescription data but do not confirm collection and ingestion of medication. Although a relatively small proportion is assumed, item numbers reported include those dispensed by doctors, resulting in an underestimation of AOM dispensed and this would apply to parental AOMs particularly. Furthermore, only the first quarter after the Covid-19 lockdown was compared to the same quarter the previous year. The use of segmented regression or ARIMA models to understand the impact of Covid-19 on subsequent quarterly time periods would have been useful.

4.6 | Future work

The ecological level observation that areas with higher prevalence of risk factors had lower rates of AOM dispensing requires confirmation using patient-level studies. These findings support research and clinical pathways that improve outcomes for osteoporosis patients, suggesting a scope for better integration of the community pharmacists in osteoporosis management. This is supported by the recommendation for research on continuity of care could promote multidisciplinary working as recommended by Royal Osteoporosis Society guidelines.⁴³

5 | CONCLUSION

The amount and type of AOM dispensed by community pharmacies varied considerably between CCGs. Populations with lower activity levels and/or a higher incidence of hip fracture should be targeted for services to address this gap in AOM provision. Community pharmacists are ideally placed to play this vital role in osteoporosis management, and further work is needed to optimise their support for effective and equitable osteoporosis management via multidisciplinary working.

AUTHOR CONTRIBUTIONS

S. S. Janjua: Conception and design, data acquisition, analysis and interpretation, writing original draft. H. F. Boardman: Data acquisition, data analysis, review and editing. A. Sami: Data analysis, review and editing. A. Johansen: Data acquisition, review and editing. L. S. Toh: Conception and design, review and editing. M. K. Javaid: Conception and design, data acquisition, analysis and interpretation, review and editing.

ACKNOWLEDGMENTS

MKJ was supported by the National Institute for Health Research (NIHR) Oxford Biomedical Research Centre (BRC). The views expressed are those of the author and not necessarily those of the NHS, the NIHR or the Department of Health.

CONFLICT OF INTEREST

MKJ has received honoraria and grant support from Amgen Ltd and UCB outside of this work. No conflicts of interest for SSJ, HFB, AS, AJ, and LST.

ORCID

Sobia S. Janjua D https://orcid.org/0000-0002-4560-0208 Helen F. Boardman D https://orcid.org/0000-0002-5341-5619 Arvind Sami D https://orcid.org/0000-0003-2961-8579 Antony Johansen D https://orcid.org/0000-0003-4726-9106 Li Shean Toh D https://orcid.org/0000-0002-9228-6266 Kassim M. Javaid D https://orcid.org/0000-0001-7985-0048

REFERENCES

- National Institute of Health. Consensus development panel on osteoporosis prevention, diagnosis and therapy. Osteoporosis prevention, diagnosis, and therapy. JAMA. 2001;285:785-795. doi:10.1001/jama. 285.6.785
- National Health Service. Accessed January 27, 2022. https://www. nhs.uk/conditions/osteoporosis
- 3. International Osteoporosis Foundation. Accessed January 27, 2022. https://www.iofbonehealth.org
- Hajbaghery MA, Abbasinia M. Quality of life of the elderly after hip fracture surgery: a case-control study. J Caring Sci. 2013;2(1):53-59.
- Panula J, Pihlajamaki H, Mattila VM, et al. Mortality and cause of death in hip fracture patients aged 65 or older - a population-based study. BMC Musculoskelet Disord. 2011;12:105. doi:10.1186/1471-2474-12-105
- Lunde A, Tell GS, Pedersen AB, et al. The role of comorbidity in mortality after hip fracture: a Nationwide Norwegian study of 38,126 women with hip fracture matched to a general-population comparison

- Leal J, Gray AM, Prieto-Alhambra D, et al. Impact of hip fracture on hospital care costs: a population-based study. *Osteoporos Int.* 2016; 27:549-558.
- Pouesmaeili F, Kamalidehghan B, Kamarehei M, et al. A comprehensive overview on osteoporosis and its risk factors. *Ther Clin Risk Manag.* 2018;14:2029-2049. doi:10.2147/TCRM.S138000
- 9. Cooper C, Campion G, Melton ⊔ 3rd. Hip fractures in the elderly: a world-wide projection. *Osteoporos Int.* 1992;2(6):285-289.
- Melton LJ 3rd, Chrischilles EA, Cooper C. Perspective how many women have osteoporosis? J Bone Miner Res. 1992;7(9):1005-1010.
- 11. Melton LJ 3rd, Atkinson EJ, O'Connor MK, et al. Bone density and fracture risk in men. J Bone Miner Res. 1998;13(12):1915-1923.
- National Health Service, Overview, Osteoporosis. Accessed January 27, 2022. https://www.nhs.uk/conditions/osteoporosis/
- Black DM, Thompson DE, Bauer DC, et al. Fracture risk reduction with alendronate in women with osteoporosis: the fracture intervention trial. FIT research group. J Clin Endocrinol. 2020;85(11):4118-4124. doi:10.1210/jcem.85.11.6953
- Yusuf AA, Cummings SR, Watts NB, et al. Real-world effectiveness of osteoporosis therapies for fracture reduction in post-menopausal women. *Arch Osteoporos.* 2018;13:33. doi:10.1007/s11657-018-0439-3
- 15. British National Formulary (BNF), 2020 BMJ Group and Pharmaceutical Press, London.
- National Institute of Clinical Excellence (NICE), Bisphosphonates for treating osteoporosis, Technology Appraisal Guidance TA464, 9 August 2017. Accessed January 27, 2022. https://www.nice.org. uk/guidance/ta464/resources/bisphosphonates-for-treatingosteoporosis-pdf-82604905556677
- 17. National Institute of Clinical Excellence (NICE), Denosumab for the prevention of osteoporotic fractures in postmenopausal women, Technology Appraisal Guidance TA204, 27 October 2010. Accessed January 27, 2022. https://www.nice.org.uk/guidance/ta204
- National Institute of Clinical Excellence (NICE), Raloxifene for the primary prevention of osteoporotic fragility fractures in postmenopausal women, Technology Appraisal Guidance TA160, Last updated: February 7, 2018. Accessed January 27, 2022. https://www.nice.org. uk/guidance/ta160
- National Institute of Clinical Excellence (NICE), Raloxifene and teriparatide for the secondary prevention of osteoporotic fragility fractures in postmenopausal women, Technology Appraisal Guidance TA161, Last updated: February 7, 2018. Accessed January 27, 2022. https://www.nice.org.uk/guidance/ta161
- Bevan H. How can we build skills to transform the healthcare system? J Res Nurs. 2010;15(2):139-148. doi:10.1177/1744987109357812
- Bhimjiyani A, Neuburger J, Jones T, Ben-Shlomo Y, Gregson CL. The effect of social deprivation on hip fracture incidence in England has not changed over 14 years: an analysis of the English hospital episodes statistics (2001-2015). Osteoporos Int. 2018;29:115-124. doi: 10.1007/s00198-017-4238-2
- Ogliari G, Lunt E, Ong T, Marshall L, Sahota O. The impact of lockdown during the COVID-19 pandemic on osteoporotic fragility fractures: an observational study. Arch Osteoporos. 2020;15:156. doi:10. 1007/s11657-020-00825-1
- McClosky EV, Harvey NC, Johansson H, et al. Global impact of COVID-19 on non-communicable disease management: descriptive analysis of access to FRAX fracture risk online tool for prevention of osteoporotic fractures. Osteoporos Int. 2021;32:39-46. doi:10.1007/ s00198-020-05542-6
- Fuggle NR, Singer A, Gill C, Patel A, Medeiros A, Mlotek AS, Pierroz DD, Halbout P, Harvey NC, Reginster JY, Cooper C, Greenspan SL (2021) How has COVID-19 affected the treatment of osteoporosis? An IOF-ESCEO Global Survey Osteoporos Int 32:611– 617. https://doi.org/10.1007/s00198-020-05793-3

WILFY⊥

[∗] WILEY-

- 25. Open Prescribing Database. EBM DataLab, University of Oxford, 2020. Accessed January 27, 2022. https://openprescribing.net
- 26. BBC News. The Changing NHS 01 Mar 2013. Accessed January 27, 2022. https://www.bbc.co.uk/news/health-19674838
- Office for National Statistics, Table SAPE22DT6a: Mid-2019 Population Estimates for 2020 Clinical Commissioning Groups (CCGs) in England by Single Year of Age and Sex - National Statistics. Accessed January 27, 2022. https://www.ons.gov.uk/peoplepopulationand community/populationandmigration/populationestimates/datasets/ clinicalcommissioninggroupmidyearpopulationestimates
- 28. Office for National Statistics, Population estimates for the UK, England and Wales, Scotland and Northern Ireland: mid-2019, using April 2020 local authority district codes. Accessed January 27, 2022. https://www.ons.gov.uk/peoplepopulationandcommunity/population andmigration/populationestimates/datasets/populationestimatesfor ukenglandandwalesscotlandandnorthernireland
- 29. National Hip Fracture database. Accessed April 30, 2021. https:// www.nhfd.co.uk
- 30. Office for National Statistics. What affects an area's healthy life expectancy? 28 June 2017. Accessed January 27, 2022. https://www. ons.gov.uk/peoplepopulationandcommunity/healthandsocialcare/ healthandlifeexpectancies/articles/whataffectsanareashealthylife expectancy/2018-01-18
- McFarlane SI, Muniyappa R, Shin JJ, et al. Osteoporosis and cardiovascular disease: brittle bones and boned arteries, is there a link? *Endocrine*. 2004;23:1-10.
- SPSS, IBM Corp. Released 2021. IBM SPSS Statistics for Windows, Version 28.0. IBM Corp
- Office for National Statistics, Table SAPE23DT6a, mid-2020, Population estimates for Super output areas, electoral wards, Parliamentary constituencies, Clinical Commissioning Groups and National Parks, released 16 Sept 2021
- National Health Service Digital, Change summary 2021 CCG mergers. Accessed January 27, 2022. https://digital.nhs.uk/services/ organisation-data-service/change-summary-reconfiguration-2021-ccgmergers
- National Institute of Clinical Excellence, Osteoporosis: assessing the risk of fragility fracture. Clinical Guideline CG146. Accessed January 27, 2022. https://www.nice.org.uk/guidance/cg146/ resources/osteoporosis-assessing-the-risk-of-fragility-fracture-pdf-35109574194373
- Hawley S, Leal J, Delmestri A, et al. Anti-osteoporosis medication prescriptions and incidence of subsequent fracture among primary hip fracture patients in England and Wales: an interrupted time-series analysis. J Bone Miner Res. 2016;31(11):2008-2015. doi:10.1002/ jbmr.2882

- 37. Pinheiro MB, Oliveira J, Bauman A, Fairhall N, Kwok W, Sherrington C. Evidence on physical activity and osteoporosis prevention for people aged 65+ years: a systematic review to inform the WHO guidelines on physical activity and sedentary behaviour. Int J Behav Nutr. 2020;17:150. doi:10.1186/s12966-020-01040-4
- Ministry of Housing, Communities & Local Government (2019) The English Indices of Deprivation. Accessed January 27, 2022. https:// assets.publishing.service.gov.uk/government/uploads/system/uploads/ attachment_data/file/835115/IoD2019_Statistical_Release.pdf
- Nyakutsikwa B, Britton J, Langley T. The effect of tobacco and alcohol consumption on poverty in the United Kingdom. Addiction. 2020; 116:150-158. doi:10.1111/add.15096
- Fone DL, Farewell DM, White J, Lyons RA, Dunstan FD. Socioeconomic patterning of excess alcohol consumption and binge drinking: a crosssectional study of multilevel associations with neighbourhood deprivation. *BMJ Open*. 2013;3:1-9. doi:10.1136/bmjopen-2012-002337
- 41. Pearson D, Taylor R, Masud T. The relationship between social deprivation, osteoporosis and falls. *Osteoporos Int*. 2004;15(2):132-138.
- 42. Tracey CA, Clarke MM, Hughes CF, et al. The impact of area based socioeconomic deprivation on osteoporosis. *Age Ageing*. 2018;47(5):v1-v12.
- National Institute of Clinical Excellence (2020). Accessed January 27, 2022. https://www.nice.org.uk/guidance/ng167/resources/ covid19-rapid-guideline-rheumatological-autoimmune-inflammatoryand-metabolic-bone-disorders-pdf-66141905788357
- Pharmaceutical Services Negotiating Committee (2021) About community pharmacy. Accessed January 27, 2022. https://psnc.org.uk/ psncs-work/about-community-pharmacy/
- Hsieh J Ecological fallacy. Encyclopedia Britannica, 4 September 2017. Accessed January 27, 2022. https://www.britannica.com/science/ ecological-fallacy

SUPPORTING INFORMATION

Additional supporting information can be found online in the Supporting Information section at the end of this article.

How to cite this article: Janjua SS, Boardman HF, Sami A, Johansen A, Toh LS, Javaid KM. Anti-osteoporosis medication dispensing by clinical commissioning groups in England – an ecological study of variability in practice and of the effect of the Covid-19 pandemic. *Pharmacoepidemiol Drug Saf.* 2022; 1-8. doi:10.1002/pds.5544